Author’s response to reviews

Title: Stimulatory effect of icariin on the proliferation of neural stem cells from rat hippocampus

Authors:
Xiaolong Fu (891981442@qq.com)
Shujun Li (hylsj001@163.com)
Shaoyu Zhou (zhoushaoy@hotmail.com)
Qin Wu (12584930@qq.com)
Feng Jin (jinfeng1115@zmc.edu.cn)
Jingshan Shi (Shijs@zmc.edu.cn)

Version: 2 Date: 07 Jan 2018

Author’s response to reviews:

BCAM-D-17-00854R1

Stimulatory effect of icariin on the proliferation of neural stem cells from rat hippocampus

Xiaolong Fu; Shaoyu Zhou; Jingshan Shi

BMC Complementary and Alternative Medicine

Dear Mr. Fu,

Your manuscript "Stimulatory effect of icariin on the proliferation of neural stem cells from rat hippocampus" (BCAM-D-17-00854R1) has been assessed by our reviewers. Based on these reports, and my own assessment as Editor, I am pleased to inform you that it is potentially acceptable for publication in BMC Complementary and Alternative Medicine, once you have carried out some essential revisions:

***********************************************************************************************

Please make the following revisions to your manuscript:

1. Please include the email addresses of all authors on the title page of your manuscript.

2. Please describe the brain extraction technique in the Methods section in more detail.
3. In your figure legends list, you have listed six figures. Please upload them as figures. We had noticed you ad uploaded figures 2-4 as supplementary files; are these separate figures, or these same figures? Please clarify.

4. Please move your Declarations to the main manuscript.

5. Please change the text under Consent to Publish to: "not applicable".

6. Thank you for including a data availability statement. This statement should detail the location of the raw data underlying the conclusions made in this study. We note that only summarized data is presented in this manuscript, so we would like to ask that you amend this statement. If there are any ethics restrictions preventing the sharing of the raw data, please state this. If you do not wish to share the data, please include a reason why it will not be shared in this section.

7. Due to the list of authors for your manuscript changing, we would appreciate it if you could please submit a change of authorship form with your revision. Please ensure to also update the list of authors on the submission system.

The authors' response letter has been included as a supplementary file